Specific Comments to Authors:
This is a very rare disease and there are many issues to learn about. However, the author needs to clarify a few issues: 1. Why was this baby born by Caesarean section instead of natural birth?

Response: Elective cesarean section because of fetal malpresentation

2. Was this child the mother’s first baby?

Response: Yes

How old was the mother?

28-year-old

Did she take any medications during pregnancy?

Response: Only prenatal multivitamin

3. The baby’s weight at birth was only 2400g. Was the baby healthy?

Response: Yes, healthy baby

Why was the baby taken to the doctor after 14 days of having jaundice?

Response: Baby was thriving well, and jaundice was thought to be physiological jaundice

4. Why wasn’t MRI performed at admission but 10 days after?

Response: The initial impression was metabolic disease

Was there any treatment within these first 10 days?
Response: No

5. The author should discuss the mother’s alloimmunity.

Response: Mother had second child healthy, she didn’t receive any prophylactic treatment for alloimmunity

6. Liver biopsy and MRI images are not consistent with PH. Did the author think this was a case of liver cell destruction due to drugs or toxins?

Response: Patient and mother was not on any medication, no history of being exposed to toxins

7. What does the author recommend to prevent this condition?

Response: Post-natal follow up visit, and earlier referral to liver transplant center.